Author's response to reviews

Title: The First Clinical Case due to AP92 like strain of Crimean-Congo Hemorrhagic Fever Virus and A Field Survey

Authors:

Kenan Midilli (kmidilli@gmail.com)
Aysen Gargili (agargili@yahoo.com)
Onder Ergonul (ondereronul@yahoo.com)
Murat Elevli (melevli@gmail.com)
Sevgi Ergin (sergin@yahoo.com)
Nesrin Turan (nturan@gmail.com)
Gönül Sengoz (drgonul@hasekihastanesi.gov.tr)
Recep Ozturk (ozturkrecep@superonline.com)
Mehmet Bakar (mbakar@gmail.com)

Version: 9 Date: 13 May 2009

Author's response to reviews:

The sentence “The study was approved by the Medical Ethics Committee of Istanbul Branch of Ministry of Health in Turkey” was inserted to the end of the Methods section.